Status
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e APU

Flight Date:

Thursday, May 12,2011
Status:

Green

All 16 APUs are currently collecting data. The units are back to charging on solar power - the hall
protection insulation was removed yesterday afternoon/evening.

All 16 APUs collected data during yesterday's event. See Dave Wolff's webpage for nice looking
images.

Notes:
APUQ9: The gauge data is likely garbage - the pulse channels on the data logger have failed.

APUO04: Wireless comms outage during the event yesterday - the data will be uploaded to the ftp
server later today.
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